Author’s response to reviews

Title: Diffusion weighted magnetic resonance imaging (DW-MRI) as a non-invasive, tissue cellularity marker to monitor cancer treatment response

Authors:

Frederikke Fliedner (frederikkepf@sund.ku.dk)
Trine Engel (tbjla@nanotech.dtu.dk)
Henrik El Ali (helali@sund.ku.dk)
Anders Hansen (aehan@nanotech.dtu.dk)
Andreas Kjaer (akjaer@sund.ku.dk)

Version: 5 Date: 05 Feb 2020

Author’s response to reviews:

Details in Funding section have been added as requested.

No other points raised in Revision.